

**SCIENTIFIC OASIS** 

# Decision Making: Applications in Management and Engineering

Journal homepage: <u>www.dmame-journal.org</u> ISSN: 2560-6018, eISSN: 2620-0104



# Application of GIS and IOT Technology-Based MCDM for Disaster Risk Management: Methods and Case Study

Nabil M. AbdelAziz<sup>1</sup>, Khalid A. Eldrandaly<sup>1</sup>, Safa Al-Saeed<sup>1,\*</sup>, Abduallah Gamal<sup>1</sup>, Mohamed Abdel-Basset<sup>1</sup>

<sup>1</sup> Faculty of Computers and Informatics, Zagazig University, Sharqiyah, Zagazig, 44519, Egypt.

ARTICLE INFO	ABSTRACT
Article history: Received 23 August 2023 Received in revised form 11 October 2023 Accepted 12 October 2023 Available online 2 November 2023 Keywords: Disaster Management; Flooding; Internet of Things, Geographic Information System; Convolutional Deep Neural Networks; Multi-Criteria Decision Making.	This study proposes a two-phase framework to enhance disaster management strategies for flooding using Geographic Information System (GIS) and Internet of Things (IoT) real-time data obtained using drones. The first phase aims to predict the governorate most prone to flooding using GIS and four forecasting models. The second phase involves selecting optimal locations for drone takeoff and landing using GIS with multi-criteria decision making. The neutrosophic ordinal priority approach is used to weight the criteria for selecting the best locations. A case study from the Egyptian Mediterranean Coast is used to measure the effectiveness and applicability of the framework. Results indicate that the Port Said governorate is the most vulnerable to flooding, and the top 10 suitable sites for drone takeoff and landing are suggested for this governorate. The limitations of the case study are discussed, such as data availability and reliability, as well as potential biases in the methodology. This study suggests future research directions to address these limitations and enhance the effectiveness of the proposed framework. Overall, this study contributes to the field of disaster risk management by providing a practical and innovative approach to enhance disaster preparedness and response using GIS and IoT technologies.

#### 1. Introduction

Disasters play an important role in affecting societies across the globe. They can cause short-term and long-lasting impacts, which alter the everyday lives of individuals and whole communities [1]. The growing severity of damages caused by disasters worldwide has accentuated the importance of disaster risk management. Disaster risk management involves the planning, coordination and implementation of strategies to mitigate the impact of disasters on human lives, infrastructure and the environment [2].

Furthermore, disaster risk management aims to develop resilience strategies against disaster events by minimising their impact [4]. The challenges in disaster management are numerous and complex. Major contributing factors, such as climate change and rapid urbanisation, are increasing

<sup>\*</sup>Corresponding author. E-mail address: <u>SEAbdelkarim@fci.zu.edu.eg</u>

https://doi.org/10.31181/dmame712024929

the frequency and intensity of disasters [5, 6]. For instance, climate change can lead to more frequent and severe weather-related disasters, such as floods, storms and heat waves. In the meantime, urbanisation can increase vulnerability to disasters by concentrating populations and assets in areas prone to hazards [7].

The management of these disasters becomes increasingly challenging due to these pervasive and escalating risks. Trends such as population growth, changing land use patterns and technological developments add further complexity [8, 9]. These factors underscore the need for continuous reevaluation and adaptation of disaster management strategies. This span of influence highlights the imperative need for effective disaster management strategies to mitigate these impacts and foster resilience. This endeavour involves forward-thinking policies and innovative technologies, such as the Internet of Things (IoT), which can provide critical data for better decision-making processes during disaster events [10].

The IoT has emerged as a powerful tool in disaster management. It facilitates real-time monitoring and data collection, which allows for early warnings and rapid response [11]. With the advent of IoT, disaster management can be revolutionised through its integration into existing systems and processes. This integration can enable the collection and analysis of vast amounts of data, which lead to more accurate predictions and timely responses [12]. One of the key benefits of an IoT-enabled disaster management system is its ability to provide real-time alerts and analyse historical data [13]. This capability enables authorities to identify areas that may require emergency services, crisis management teams and critical relief in the event of a disaster. Furthermore, integrating IoT into smart cities is essential for enhancing disaster management capabilities. By leveraging IoT technologies, smart cities can optimise their disaster management facilities and reduce the harmful impact of disasters [14]. For example, by utilising IoT in smart city applications, the detection of disasters such as building fires, atmospheric pollution and route blockages can be achieved more efficiently and in a timely manner [15, 16]. In addition to improving response capabilities, IoT can play a vital role in the prevention and mitigation of disasters. IoT sensors can be deployed to monitor environmental conditions, such as humidity levels, temperature variations and water levels, for detecting potential hazards and taking proactive measures to prevent disasters or minimise their impacts [17]. Moreover, IoT can enable more effective coordination and communication during rescue events [18]. Therefore, IoT technologies can facilitate seamless communication and collaboration amongst different stakeholders involved in disaster response, including emergency services, government agencies, NGOs and volunteer organisations [19]. In conclusion, the integration of IoT in disaster management offers numerous benefits and opportunities for improving response capabilities, prevention and mitigation efforts [20].

In the field of disaster management, Geographic Information Systems (GIS) have proven to be invaluable tools for data collection and analysis. These systems allow for the visualisation, mapping and spatial analysis of data related to disaster events. By incorporating IoT technologies into GIS, disaster management practitioners can gain access to real-time data that can aid in decision-making processes. GIS complements IoT by providing a spatial context to the collected data. This way allows for the visualisation and analysis of the data on a map, which assist in the identification of high-risk areas, planning evacuation routes and allocating resources effectively [21, 22]. Intelligent multiobjective optimisation (IMOO) techniques also play a crucial role in disaster management. These techniques involve optimising multiple objectives simultaneously, such as minimising response time, maximising resource allocation efficiency and reducing the impact of disasters; this feature helps decision makers make informed decisions and allocate resources efficiently in real time during crisis situations [23–25]. Moreover, convolutional deep neural networks (CDNNs) are emerging as

powerful tools in disaster management [26, 27]. They can process large amounts of data, including satellite imagery and social media data, to identify patterns and trends that can aid in disaster response efforts [28]. These technologies, when used together, can considerably enhance the capacity to expedite relief efforts during any emergency or disaster.

The research gap addressed by this study is the lack of effective IoT data collection strategies for disaster risk management, particularly in flood-prone regions. This study proposes the use of drones equipped with sensors capable of gathering and transmitting data in real time before, during and after disasters. This study also identifies the need for a comprehensive evaluation of criteria and subcriteria in the multi-criteria decision making (MCDM) model to determine optimal locations for IoT data collection. Overall, a framework for utilising IoT and GIS technologies in disaster risk management is provided, which fills a research gap in the field.

# 1.1 Objectives of the Research

- 1 To explore the potential of IoT and GIS technologies in enhancing disaster risk management.
- 2 To predict the region most prone to flooding using GIS and forecasting models.
- 3 To identify the optimal locations for drone takeoff and landing using GIS with MCDM for maximising the efficiency and effectiveness of data collection.
- 4 To apply a neutrosophic-based MCDM method called neutrosophic ordinal priority approach (N-OPA) for weighting the criteria and sub-criteria to select the optimal locations.
- 5 To employ a case study from the Egyptian Mediterranean Coast for assessing the effectiveness and applicability of the framework in enhancing disaster management strategies.

# 1.2 Contributions of the Research

- This study improves disaster management strategies, particularly drone strategies in flood-prone areas along the Egyptian Mediterranean Coast. Precise determinations of flood-prone governorates are provided, which allows for targeted resource allocation and mitigation strategies.
- Traditional prediction models (curve fit forecast, exponential smoothing and forest-based forecast) are compared with the CDNN model (InceptionTime) to identify flood-prone governorates.
- The development of an MCDM model contributes to the academic literature, specifically in optimising the selection of locations for drone takeoff and landing, which is a novel aspect of disaster management.
- This study offers a remarkable contribution to data collection and processing methods by demonstrating effective approaches for gathering and analysing data in disaster management. A notable contribution of this study is the application of the N-OPA method to determine weights of criteria and sub-criteria in a decision-making context, which provides practical guidance for other researchers and practitioners.
- The application of GIS processing assists in identifying suitable locations for drone takeoff and landing, which represents an innovative approach in disaster management. This method could potentially inspire further studies or applications in similar contexts.

This study contributes to addressing research gaps in disaster risk management by proposing a two-phase framework that utilises IoT and GIS technologies to enhance disaster management strategies in flood-prone regions. Specifically, this study fills the gap in effective IoT data collection strategies by proposing the use of drones equipped with sensors capable of gathering and

transmitting data. These drones can access inaccessible and unsafe areas for humans, which provides real-time data before, during and after disasters.

This study also employs a comprehensive evaluation of criteria and sub-criteria in the MCDM model to determine the best sites for drone takeoff and landing, which fills a research gap in selecting optimal locations for IoT data collection. The use of a neutrosophic-based MCDM model called N-OPA handles uncertainty in decision making, which is a crucial aspect in disaster management.

Furthermore, this study provides practical insights into the application of IoT and GIS technologies in disaster risk management through a case study in the Egyptian Mediterranean Coast region. The findings of this study can be utilised by policymakers, disaster management authorities and other stakeholders to develop better planning and decision-making strategies for minimising the risk and mitigating the impact of natural disasters.

### 1.3 Problem Statement and Research Questions

This study aims to address the challenge of effectively using IoT and GIS technologies in disaster risk management, specifically in predicting flood-prone regions and selecting optimal sites for drone takeoff and landing. The goal is to maximise the efficiency and effectiveness of data collection during and after disasters. The research statement identifies the problem of inadequate disaster risk management in flood-prone regions and highlights the usefulness of emerging technologies such as GIS, IoT, forecasting models and MCDM in addressing this issue. This statement is important because it sets the stage for the rest of the study, which guides the reader's understanding of the problem and the proposed solutions and helps them engage with the findings. It also provides clarity and direction for future researchers who may wish to explore similar topics.

Based on the problem statement of this study, the following research questions are identified:

- How can IoT and GIS technologies be effectively used in disaster risk management?
- How can flood-prone regions be accurately predicted using these technologies?
- How can optimal sites for drone takeoff and landing be identified for data collection during and after disasters?
- How can the efficiency and effectiveness of data collection be maximised through the use of these technologies?

# 1.4 Organisation of the Paper

The rest of the paper is structured as follows. Section 2 presents an overview of previous research related to IoT and supporting techniques in disaster management. Section 3 discusses the research framework and the methods used. Section 4 presents the case study, which illustrates a real-world application in disaster management. Section 5 introduces the discussion and conclusion.

### 2. Literature Review

### 2.1 IoT for Disaster Management

An IoT system refers to a network of interconnected devices, sensors and software applications that collect and exchange data through the Internet. These devices, sensors and applications are embedded in various physical objects, such as buildings, vehicles and infrastructure [29, 30].

IoT technologies and applications in disaster management scenarios greatly improve response times, increase situational awareness and enhance overall disaster management capabilities. IoT technologies related to disaster management facilitate real-time monitoring, data collection and analysis. This way enables stakeholders to gain a comprehensive understanding of the situation on the ground and make informed decisions. For example, an IoT-enabled disaster management system can provide real-time alerts and collect historical data to determine areas in need of emergency services and critical relief during a disaster.

The IoT system for disaster management includes three main components, namely, edge, network and core, as shown in Figure 1.

- **IoT Edge**: The IoT edge refers to the devices and sensors that are deployed at the periphery of the network, close to where data are collected [31].
- **IoT Network**: The IoT network is the communication infrastructure that links these devices and sensors to the central processing and storage systems, which enables real-time decision making.
- **IoT Core**: The IoT core is a managed cloud service that handles secure connection, management and data processing from IoT devices. It simplifies device deployment and management whilst facilitating real-time data analytics in disaster management.





Fig. 1. IoT system components

# 2.2 IoT Applications in Disaster Management

IoT has various applications in disaster management, which can be categorised into four main phases: mitigation, preparedness, response and recovery. Mitigation: IoT helps minimise theimpact of disasters by predicting and preventing incidents before they occur. Examples of IoT applications in mitigation include [32–34] monitoring environmental changes and predicting natural disasters like forest fires, floods and landslides. IoT also identifies weak infrastructure and structural flaws in buildings that may be vulnerable to natural disasters. Preparedness: IoT devices assist in effective planning and communication during disasters. Applications of IoT in preparedness include creating early warning systems to alert communities about impending disasters [35], enhancing communication networks for first responders and emergency management agencies and developing real-time monitoring systems to track the availability of critical resources like food, water and medical supplies [36, 37]. Response: IoT devices play a crucial role in managing disaster response operations by providing accurate real-time data. Applications of IoT in disaster response include deploying unmanned aerial vehicles (UAVs), which are aircraft operated without a human pilot on board. These vehicles are often equipped with various sensors, cameras and other technologies that allow them to perform various tasks. For example, during flooding, real time data collected by UAVs can be used to gather information, assess damage and locate survivors [38]. This feature enables smart tracking of rescue teams and equipment to optimise response times and resource allocation [39] and facilitates remote medical consultations and telemedicine during disasters. Recovery: IoT technology is instrumental in efficient recovery and rebuilding efforts after a disaster.

Some applications of IoT in recovery include [40] assessing the extent of damage using remote sensing and AI-based image analysis, managing the distribution of critical resources and supplies to affected areas based on real-time data, monitoring and analysing post-disaster environmental conditions to ensure safety for recovery efforts and implementing IoT-enabled systems to automate and streamline recovery processes, such as debris removal and reconstruction.

### 2.3 Challenges and Limitations of IoT for Disaster Management

Despite the tremendous potential of IoT in disaster management, several challenges and limitations need to be addressed to ensure its effective implementation [41, 42]:

- **Connectivity and network issues**: Disasters often result in infrastructure damage, which leads to disrupted communication networks. Reliable connectivity is essential for IoT devices to operate efficiently, and network failures can severely hamper real-time data transmission and coordination efforts.
- **Power supply:** Ensuring a continuous power supply to IoT devices during disasters can be challenging, which is due to that power outages are common in disaster-hit areas. Developing energy-efficient devices and alternative power sources like solar panels or batteries is crucial to address this concern.
- Scalability concerns: The rapid deployment of IoT devices in large numbers can pose challenges to scale, especially in managing, monitoring and processing vast amounts of data generated during disasters.
- **Data security and privacy issues**: The sensitive nature of data collected and shared during disasters raises concerns on data security and privacy. Ensuring that data are protected from unauthorised access and tampering is important to maintain trust and confidentiality.
- Interoperability: Efficient disaster management requires seamless integration of IoT devices with various platforms, agencies and systems. IoT devices from different manufacturers may have differing protocols and standards, which makes interoperability a primary challenge.
- Limited resources and budgets: Many disaster-prone areas, particularly in developing regions, have limited resources and budgets. The cost of implementing and maintaining complex IoT-based disaster management systems can be prohibitive for these regions. Addressing these challenges requires a holistic approach that combines IoT with other technologies and strategies.

# 2.4 Supporting Techniques for Disaster Management

- GIS: GIS is a technology that enables the capture, analysis and visualisation of geospatial data [43–45]. GIS is an essential tool in disaster management because it can be used to map areas prone to disasters, analyse their vulnerability and identify critical infrastructures that may be affected [46, 47]. GIS can also be used to monitor the spread of disasters and support search and rescue operations by providing real-time updates on affected areas and routes for evacuation [48]. Moreover, the use of GIS facilitates effective mapping and visualisation of disaster-affected areas, which provides decision makers with vital information for resource allocation [49, 50].
- CDNN: CDNNs are a type of artificial neural network optimised for image processing tasks, such as object recognition and classification [51]. In the context of disaster management, CDNNs can be used to analyse satellite and drone imagery for damage assessment in affected areas. Furthermore, CDNNs in disaster management assist in identifying critical infrastructures that may be affected by disasters and support decision making on resource allocation and rescue operations [52–56].
- MCDM: MCDM is an analytical approach designed to consider multiple, often conflicting, criteria in decision-making processes in the context of disaster management [57, 58]. It provides a structured, objective methodology for prioritising or ranking various options in response to a disaster scenario [59]. MCDM helps in evaluating different disaster management strategies based on multiple criteria, such as cost, effectiveness, feasibility and impact on society. By considering multiple criteria, MCDM provides a comprehensive view of potential outcomes of different strategies and helps decision makers choose the best course of action [60, 61]. This approach is particularly useful in disaster scenarios where decisions often need to be made quickly, under high stress and with limited resources. MCDM offers a rational, flexible and systematic approach adaptable to various types of disasters and disaster management scenarios [62].
- **IMOO**: IMOO techniques refer to a set of algorithms and methods that can optimise multiple conflicting objectives simultaneously whilst considering constraints [63]. In the context of disaster management, IMOO techniques can be used to optimise resource allocation during response operations [64, 65]. For example, they can help in determining the optimal distribution of relief supplies to maximise their impact and minimise wastage [66]. IMOO techniques can also assist in decision making under uncertainty by generating multiple solutions to a problem, which allows stakeholders to choose the best option based on their specific needs and preferences [67]. These techniques have also been utilised to optimise resource allocation and decision making during disaster response.

### 2.5 Applications of Integrating IoT with Supporting Techniques for Improved Disaster Management

The integration of IoT, GIS, MCDM, IMOO and CDNNs can highly improve disaster management in various ways and can be applied to various aspects of disaster management. **Real-time Monitoring and Quick Response**: IoT devices, when deployed in disaster-prone areas, facilitate real-time monitoring. They can promptly detect changes in environmental conditions and send alerts. Through MCDM, decisions regarding rapid response, considering various conflicting criteria, can be made efficiently. **Geospatial Analysis and Prediction**: A collaboration of IoT and GIS can offer considerable geospatial analysis benefits. Sensors and satellite data can be visualised over geographical scales to better understand disaster patterns and predict vulnerable areas. Efficient Resource Allocation: Applications of IMOO in disaster management involve optimising limited resources during emergencies. The techniques focus on meeting multiple objectives like minimising casualties and ensuring prompt responses. **Data-Driven Decision Making and Predictive Modelling**: IoT provides a large amount of real-time data that can be analysed through CDNNs. CDNN algorithms process the data, identify patterns and deliver crucial insights for better decision making. These neural networks can also predict potentially disastrous scenarios, which enables early mitigation measures. **Improvement in Mitigation Strategies**: The integration of MCDM with other technologies enhances the decision-making process, especially when various performance criteria are involved. In this way, mitigation strategies can be continuously evaluated and improved. **Assistance in Post-Disaster Recovery**: After a disaster, large amounts of data collected by IoT devices can help in effective relief operations, damage assessment and rehabilitation efforts. **Enhanced Communication**: The collaboration of these technologies can improve information sharing and communication amongst different stakeholders involved in disaster management.

### 2.6 Challenges and Potential Solutions of Integrating IoT with Supporting Techniques for Improved Disaster Management

Challenges include the following: **Technical Challenges**: The integration of IoT, GIS, MCDM, IMOO and CDNN requires complex programming and data management skills. The ability to capture, process, store and analyse vast amounts of data in real time can be challenging. **Data Privacy and Security**: Given the sensitivity of data gathered during disaster management, ensuring data security from malicious entities is important. The large datasets involved may also raise privacy and data ownership issues. **Interoperability**: Each technology (IoT, GIS, MCDM, IMOO and CDNN) uses specific protocols and standards that may not be compatible with one another. This mismatch can lead to limited communication and functionality amongst integrated systems. **Infrastructure**: The effective operation of these technologies requires a reliable and robust infrastructure. In disaster-prone areas, consistent power supply and connectivity might be challenging.

Potential solutions include the following: **Training and Capacity Building**: These technologies can be effectively managed by investing in technical teams and enhancing their skills. Capacity-building opportunities can be provided for data scientists, network engineers, disaster management professionals and other related personnel. **Implementing Data Security Measures**: Data encryption, strict data access rules and secure cloud storage options can be used to protect data privacy and security. **Standardisation**: The interoperability issue can be addressed by standardising data formats and communication protocols. This solution would likely require coordination amongst technology manufacturers, academics, and disaster management authorities. **Robust Infrastructure**: Investment in reliable and resilient infrastructure can be considered, which can include the use of solar power or other forms of renewable energy to ensure a consistent power supply and the construction of hardened data centres that can withstand adverse conditions.

# 2.7 Future Outlook and Research Gaps of Integrating IoT with Supporting Techniques for Improved Disaster Management

Future outlooks and directions may focus on the following: **Scale and Scope**: The application of IoT, GIS, MCDM, IMOO and CDNN technologies in disaster management will widen in the future. As these technologies mature, their integration will become even more seamless and impactful, which facilitate a comprehensive, real-time understanding of disaster scenarios. **Predictive Capabilities**: Enhancements in AI and machine learning capabilities will likely improve the predictive capabilities of these technologies, which enhances risk assessment, early warning and disaster response efforts. **Smart Cities**: As more cities become 'smart', the integration of these technologies within urban

planning and infrastructure will be critical in improving their resiliency against disasters. Research gaps include the following: **Human Factors**: Current research lacks comprehensive studies on human factors in technological adoption. Although these technologies offer vast possibilities, the implementation, usage and acceptance by end-users, especially disaster management professionals, need more attention. **Ethical and Privacy Considerations**: Future research should focus on ethical guidelines and methods to protect sensitive data collected during disaster management. **Case Studies**: More case studies are needed to demonstrate the practical application of integrating IoT, GIS, MCDM, IMOO and CDNN in real-world disaster scenarios. **Efficacy Studies**: Research is needed to effectively quantify and compare the benefits of this technological integration over traditional disaster management methods. **Interoperability**: A knowledge gap exists in understanding the effective integration of these different technologies and ensuring their efficient communication to enhance disaster management.

Existing studies have largely explored the individual applications of IoT and GIS technologies in disaster risk management. IoT has been used for real-time data collection and analysis in disaster response, whilst GIS has been utilised for spatial analysis and mapping of disaster-prone areas. However, few works have investigated the integration of these technologies to enhance disaster risk management strategies. Recent study proposed an integrated framework combining IoT and GIS to improve flood monitoring and early warning systems [68]. The framework utilised IoT sensors to collect hydrological and meteorological data, which were then analysed using GIS for flood modelling and mapping. A similar framework were used to predict urban flash floods using machine learning algorithms and IoT sensors [69]. These studies demonstrate the potential of integrating IoT and GIS technologies in disaster risk management. However, a gap in the literature still exists in the selection of optimal drone site locations for data collection in disaster management. Our research aims to address this gap by utilising MCDM for prioritising criteria for drone site selection and considering uncertainty in decision making using a neutrosophic approach. Neutrosophic MCDM is well suited for our research because it allows for the consideration of uncertain and incomplete information in decision making. The use of neutrosophic sets, which account for uncertainty and indeterminacy, can provide a more accurate and comprehensive representation of complex decision-making criteria in disaster management. This approach is particularly important when dealing with multiple competing criteria and uncertainty in site selection for drone deployment during disasters. Overall, previous studies have analysed the individual benefits of IoT and GIS technologies in disaster risk management, but fewer works have integrated these technologies and utilised MCDM to optimise drone site locations for data collection. Our research aims to address this gap in the literature and utilise a neutrosophic approach to account for uncertainty in decision making.

#### 3. Research Framework

This study takes advantage of integrating techniques and aims to predict which governorate across the Egyptian Mediterranean Coast is most susceptible to flooding in the next 5 years using GIS and CDNNs. Then, MCDM is used with GIS to identify the most suitable locations for drone takeoff and landing to monitor the flood-prone area for an early warning system. This study consists of a two-phase framework with five main steps, as illustrated in Figure 2. Phase one comprises the first step, where the primary aim is to accurately determine the flood-prone governorates along the Egyptian Mediterranean Coast. Phase two consists of steps 2–5. In Step 2, an MCDM model is defined. The MCDM model incorporates various criteria and attributes relevant to selecting the most suitable locations for drone takeoff and landing. Step 3 involves data collection and processing. In Step 4, the

weights of criteria and sub-criteria are determined using the N-OPA method. In Step 5, GIS processing is conducted to identify suitable locations for drone takeoff and landing.





# 3.1 Study Area

The study area includes the governorates across the Egyptian Mediterranean Coast, as illustrated in Figure 3. This geographical region is particularly relevant to the study due to its susceptibility to flooding and its need for improved disaster management strategies [70–73]. By implementing the research framework within this context, the study aims to demonstrate the integration and effectiveness of GIS, MCDM and CDNN technologies in enhancing disaster preparedness and response.



Fig. 3. Study area

This goal is achieved through accurate predictions of flood-prone areas and the identification of suitable areas for drone deployment in emergency situations. The governorates across the Egyptian Mediterranean Coast include the following: Alexandria: It is the second-largest city in Egypt and is renowned for its historical landmarks. Port Said: This coastal governorate, which is situated along the Mediterranean, is known for its important economic and commercial activities due to the Suez Canal. Damietta: This region is famous for its furniture industry and is located at the Damietta branch of the Nile River. Beheira: It is located in the northern part of the country, and it has an extensive coastline along the Mediterranean Sea. Kafr El Sheikh: This governate is recognised for its agriculture, especially riceand fish farming. Matrouh: It is located on the western edge of Egypt along the Mediterranean coast, and it is renowned for its beautiful beaches. North Sinai: This region holds strategic importance due to its proximity to Israel and the Gaza Strip. Dakahlyia: This governorate of Egypt is located in the northeastern part of the country in the Nile Delta region. Its capital is Mansoura. It does not border the Mediterranean coast, but it is in close proximity and greatly influences the coastal region. It susceptible to the impacts of disasters, such as floods, due to its geographical location and dense population. Thus, it is a relevant area for disaster management studies. Each of these governorates has unique characteristics that can impact the application and effectiveness of IoT and its supporting techniques in disaster management.

#### 3.2 Panel of Experts

The panel of experts for this study comprises professionals specialised in the fields relevant to the research. It includes representatives from the following disciplines:

- Disaster Management Professional (E1): An individual with experience and knowledge in disaster response, emergency planning and management strategies.
- Data Scientist (E2): An expert with the ability to process and analyse large datasets and a solid background in AI and machine learning methods, especially CDNNs.
- GIS specialist (E3): An expert with experience in GIS, who provides valuable insights into spatial data collection, processing and interpretation.
- Decision Science Specialist (E4): An expert who can assist in structuring MCDM model and validating the weighting and scoring process.
- Urban Planner (E5): A professional with knowledge of the infrastructure and unique characteristics of the study area, which includes the governorates across the Egyptian Mediterranean Coast.

# 3.3 Data Collection and Processing

A comprehensive data collection process is necessary to conduct research described in the previous steps. For the prediction phase of the research framework, historical data were collected from the website for accessing data from the National Aeronautics and Space Administration (NASA). The data and variables used for the prediction included rainfall and humidity historical data for 20 years, which were collected on an hourly basis from March 2003 to March 2023. To address missing values in the dataset, the approach of dropping locations (or observations) with missing data was used. This approach has several advantages:

- **Simplification**: It simplifies data handling given that advanced techniques like data imputation can be complex or may introduce bias in some cases.
- Larger Datasets: When dealing with large datasets, dropping records with missing values does not significantly impact statistical power.
- Accuracy: By dropping locations with missing data, the accuracy of predictions can be improved because incomplete or unreliable data are excluded from the analysis.

The collected datasets were converted to spatial layers (point shape files) using ArcGIS Pro 3.1 software. Then, the 'Create Space Time Cubes' (from defined locations) tool was used to examine data in space and time. This tool generates cube bins, with each bin representing a weather station for a particular 1 h time period. Each column in the cube corresponds to a location, with each location representing a weather station. The 'Visualise Space Time Cube in 3D' tool was used to visualise the created space time cubes. The MCDM model was built to find the best areas for drone takeoff and landing. The second part of the study framework involves collecting and processing data to support the MCDM model, as shown in Figure 4. Spatial data were collected from secondary sources, including the global digital elevation model (DEM) produced by NASA, data on roads and coastline from the Egyptian National Authority for Remote Sensing and Space Sciences (NARSS), information about airports and population from the Egyptian Central Agency for Public Mobilisation and Statistics (CAPMAS) and air quality index data from the Breezometer website. In addition, data such as slope, restricted areas and land use were generated using various analytic tools found in ArcGIS Pro 3.1 Pro software.



Fig. 4. Criteria and sub-criteria used in the MCDM model

Further details of the criteria and sub-criteria of the MCDM model are shown in Table 1. After data collection, standardisation was performed. This step is crucial in MCDM methods to ensure that all the criteria are on a common scale, which prevents any single criterion from dominating the decision-making process due to differences in numerical units. Standardisation ensures that each criterion contributes equally to the decision-making process, which facilitates accurate comparisons and effective decision making. The min–max normalisation method, which is one of the most commonly used standardisation methods, was used. This approach rescales the values of each criterion to a range of 0 to 1 [74]. It is achieved using the formula (x-min) / (max-min), where x is the original value, max is the maximum value and min is the minimum value. Following standardisation, the decision maker can effectively compare and aggregate the different criteria to arrive at a final decision.

#### Table 1

Description of the evaluation criteria and sub-criteria used in the MCDM model

Criteria	Description	Sub-criteria	Description
C1: Legal It refers to all the legislation an		A11: Distance from Airport	It refers to the physical distance of the potential drone operation site from the nearest airport.
Regulations	rules that must be adhered to when operating drones.	A12: Restricted Areas	It refers to zones where drone operations may be limited or prohibited due to various reasons.
C2: Accessibility	It refers to the ease of reaching the selected sites for drone takeoff and landing. It can be crucial for regular	A21: Accessibility to Water Bodies	It signifies the level of accessibility for a drone to reach

Criteria	Description	Sub-criteria	Description
	maintenance, emergencies, and equipment replacement purposes.		bodies of water from the selected location.
		A22: Distance from Roads	It refers to the proximity of a selected location to the nearest road infrastructure.
		A31: Elevation	It refers to the height above a fixed reference point, which is typically defined as sea level.
C3: Topography	It refers to the physical features of the selected locations. It consists of the detailed study of the surface features of the land.	A32: Slope	It refers to the degree of steepness or inclination of a surface.
	features of the land.	A33: Distance from Drainage	It refers to the separation distance between a particular location and the nearest drainage features, such as a river, stream or artificial drainage system.
	It refers to the suitability of the	A41: Proximity to Populated Areas	It refers to the closeness or distance to regions with significant human populations.
C4: Safety	without posing a risk to property or individuals or the drone itself.	A42: Obstruction- free Area	It refers to spaces that are free from physical barriers or impediments that might hinder drone operations.
C5:	It refers to the climatic and	A51: Noise Limitations	It refers to restrictions or regulations on acceptable levels of noise in certain areas due to drones.
Environmental Conditions	It refers to the physical features of the selected locations. It consists of the detailed study of the surface features of the land. A33: Distance from Drainage A41: Proximity to Populated Areas A42: Obstruction- free Area It refers to the climatic and environmental factors of the potential sites for drone takeoff and landing. A52: Air Quality	It refers to the condition of the air within a specific region or the overall atmospheric conditions. It is influenced by natural and man- made factors.	

### 3.4 Forecasting Models

### 3.4.1 InceptionTime Model

InceptionTime is a type of CDNN that is specifically designed for time series data. It draws inspiration from the more general Inception network used for image processing but has been adapted to handle sequential data [75]. Each layer of InceptionTime consists of parallel branches of convolutional layers with different kernel sizes before their results are concatenated. This design enables the model to capture patterns of different time intervals.

InceptionTime essentially serves as a CDNN model designed for time series classification [76]. Its structural foundation is rooted in Google's Inception module for image classification. The key steps in InceptionTime include the following [77]:

- **Reshape the Dataset**: Time series data are usually reshaped into a format suitable for convolutional processing. If the dataset is a univariate time series sample of length n, then it is reshaped to (n, 1).
- **Construct Inception Module**: InceptionTime uses an inception module, which is a small collection of parallel convolutional layers with different kernel sizes. This module captures temporal patterns of different time frames.

- **Create Convolutional Blocks**: The inception modules are stacked together multiple times to create several convolutional blocks. Each block can capture multi-level abstractions of the data.
- **Global Average Pooling**: After passing the time series through the Inception modules, a global average pooling layer is applied. This way reduces the output to a 1D vector, which retains only the most significant features and reduces overfitting.
- Fully Connected Layer and Softmax Function: The output from the global average pooling undergoes processing through a fully connected layer, followed by a softmax function for multi-class classification.
- **Model Training**: The model is trained using backpropagation and an optimisation algorithm.
- **Prediction**: Following training, the model can classify new, unseen time series data.

InceptionTime is an appropriate method for this study for several reasons: Handling Time Series Data: InceptionTime is designed to handle time series data, which makes it suitable for this study that presumably involves chronological data, such as weather patterns or disaster incident reports. Scalability: InceptionTime can process large amounts of data without a significant decrease in speed or an increase in required computational resources. This scalability is important in disaster management, where extensive datasets are common. Detecting complicated Patterns: The InceptionTime model can detect complicated temporal patterns within the input by employing multiple convolutional layers with various kernel sizes concurrently. This capability helps in capturing short- and long-term dependencies, which is crucial in disaster management. Robustness: This model demonstrates robustness to noise and missing values in the dataset, which are common occurrences in disaster management data. Reduced Risk of Overfitting: The use of global average pooling, which handles regularisation automatically, reduces the risk of overfitting. Ultimately, the potential performance of the model on unseen data improves. Predictive Performance: InceptionTime has demonstrated high predictive performance on a wide range of tasks, which provides a solid basis for its application in this study.

# 3.4.2 Curve Fit Forecast Model

Curve fitting is a type of regression analysis used to find the 'best fit' curve or line for a series of data points. It can be used for forecasting, explaining variations in data or detecting anomalies.

Curve fitting involves the following steps [78]:

- 1 Data Collection: The relevant data points to be plotted on the curve are collected.
- 2 **Select an Appropriate Model**: A mathematical model or function that can best describe the data is selected. The function may be a simple line (linear regression) or a complex curve (polynomial regression). The model selection depends on the nature of the data and the purpose of the analysis.
- 3 **Estimate Parameters**: The values of the parameters in the model function that produce the best fit to the data are determined. This procedure is typically conducted by minimising the sum of the squared differences between observed and predicted values (using the least square method).
- 4 **Validate the Model**: The accuracy of the model is tested by predicting the values of the dependent variable and comparing them with the actual values.

Curve fitting can be suitable for this study for the following reasons: **Simplicity andFlexibility:** Curve fitting is a simple and flexible way to model complexrelationships between variables. Its simplicity makes it easy to implement and interpret. **Accurate Forecasts**: If the selected model closely matches the pattern of the data, then curve fitting can produce highly accurate forecasts. **Quantifying Uncertainty:** Researchers can determine the degree of data variability around a best-fit curve through its generation. This feature helps quantify the uncertainty in predictions and aids in developing robust disaster mitigation strategies. **Detecting Trends:** Curve fitting enables researchers to identify and analyse trends in the data. This capability can be especially useful for understanding patterns in disaster incidents, forecasting and planning effective responses.

# 3.4.3 Exponential Smoothing Model

Exponential smoothing is a time series forecasting method for univariate data that involves the application of decreasing weights for past observations, with the weights decreasing exponentially. The method is popular due to its simplicity, efficiency and low computational resource requirements. The steps involved in this method are as follows:

- 1 **Choose Smoothing Constants:** An initial smoothing constant is selected, which is usually represented by the Greek letter alpha ( $\alpha$ ). The value of  $\alpha$  is between 0 and 1, with values closer to 0 placing more weight to older data and values closer to 1 emphasising more recent data.
- 2 **Calculate Simple Moving Average**: A simple moving average is calculated as the initial forecast. This procedure can be conducted by averaging the first few data points in the series.
- 3 **Apply Exponential Smoothing**: To obtain the next forecasted value,  $\alpha$  is multiplied by the actual value of the present time period. Then, it is added to the quantity (1 minus  $\alpha$ ) multiplied by the forecasted value in the previous period.
- 4 **Repeat the Process**: The process is repeated by using the new forecasted value as the previous forecasted value for the next calculation.

Exponential smoothing is suitable for this study because of several key reasons: Adaptability: It works effectively with datasets showing stable patterns over time and can adapt to changes in level, trend and seasonality in the data. Efficiency: It is computationally efficient, which enables immediate forecasting as soon as the latest data become available. Simplicity: It is relatively easy to understand, apply and explain to others, which make it an attractive choice in many forecasting situations. Intuitive: The exponential decay provides an intuitive way to 'weigh' the data, with more recent data having more influence on the smoothing process. When using exponential smoothing, model assumptions and the quality of the fit need to be checked regularly as more data become available to confirm that the statistical properties remain stable.

# 3.4.4 Forest-Based Forecast Model

Forest-based forecasting is a machine learning technique that utilises multiple decision tree models, which work as an ensemble to generate a final forecast. The 'forest' consists of individual 'trees', with each tree being unique and trained on different subsets of the original data. The final forecast is a combination of these individual forecasts. The steps involved in forest-based forecasting include the following [79]:

- 1 **Data Preparation**: The initial dataset is divided into multiple subsets, which often involves bootstrapping or random sampling with replacement.
- 2 **Building Trees**: Each subset is used to grow a decision tree. The tree is created by making several decisions or splits based on criteria such as reducing the variance in forecast error. Each tree is grown fully and is not pruned, as in a single decision tree model.
- 3 **Making Individual Forecasts**: After all the trees are grown, a new data point is passed through each tree to reach a forecast.

4 **Aggregating Forecasts**: The final step in a random forest is to calculate an average (for regression problems) or perform a majority vote (for classification problems) based on the results from all trees for each new data point.

Forest-based forecast approach is appropriate for this study because of its capabilities. **Handling Complex Relations:** Forest-based forecasting can capture complex nonlinear relationships between predictors and the variable to be forecasted. This capability is crucial for achieving accurate predictions in disaster management. **Robust**: It exhibits robustness to outliers and does not require extensive preprocessing and scaling. This feature is advantageous when dealing with real-world, messy data. **Multiple Variables**: This method can simultaneously handle multiple input variables, which allows for the consideration of various relevant factors, such as meteorological variables, in the prediction. **Interactions**: Random forests automatically model interactions between variables, which can be essential in this context.

### 3.5 N-OPA

The neutrosophic environment is a mathematical framework primarily based on neutrosophic set theory, which is an extension of fuzzy set theory. Neutrosophic set was proposed by Florentin Smarandache in 1995 to handle simultaneous indeterminate and inconsistent information [80]. It introduces three membership degrees: truth (T), indeterminacy (I) and falsity (F) [81], each of which independently ranges between 0 and 1. This method differs from previous models, which often consider true and false as complementary. The N-OPA is an element of neutrosophic MCDM. It focuses on decision-making processes where the available data are imprecise, undefined or uncertain. In a typical decision-making process, options are ranked based on a precise quantification of their qualities. However, this precise quantification is not always possible, which leads to potential inaccuracies. The N-OPA, instead, allows these qualities to be ranked in an ordinal manner, from 'most preferred' to 'least preferred', without needing defined numerical values of each quality. N-OPA uses neutrosophic numbers, rather than traditional ones. These neutrosophic numbers hold three independent values: truth (T), indeterminacy (I) and falsity (F). In practical terms, this way allows decision makers to consider not only the potential benefits (T) and drawbacks (F) of a decision but also the level of uncertainty or ambiguity (I) associated with it. For a real-time proposed approach like the N-OPA, the benefit lies in the ability to analyse and make decisions under indeterminate and inconsistent information. When fed with real-time data, N-OPA can quantify and encapsulate the uncertainty inherent in such data using the indeterminacy membership degree (I). By factoring in this degree whilst making decisions, N-OPA can provide outputs that are more reflective of the intricacies and uncertainties present in real-world data. This decision-making method has its advantages, such as the capacity to address complicated and vague problems or those with incomplete information. N-OPA can be a practical tool for decision makers facing complex uncertain scenarios.

### 3.5.1 Advantages of N-OPA

- Handling Ambiguous and Inconsistent Data: In real-life scenarios, data are frequently inconsistent or lacking. Such information can be used by N-OPA to produce accurate findings. N-OPA uses neutrosophic numbers to address ambiguities, which enhances its reliability in ambiguous contexts. It can handle para consistent information, which classical decision-making methods may struggle with.
- Expressing Confidence and Uncertainty: N-OPA allows decision makers to express their degree of confidence and uncertainty in their judgments. This feature is particularly valuable for capturing the complexity of decision-making situations.

- Representing Contradictions: Neutrosophic sets allow for the representation of contradictions, which can be useful in resolving conflicting criteria in decision-making situations.
- Systematic and Quantitative Decision Making: N-OPA provides stakeholders with a systematic and quantitative means to consider different criteria. This way improves clarity, reduces bias and allows for the weighting of criteria using their importance and non-importance. Ultimately, multiple perspectives are considered in decision making.

# 3.5.2 Disadvantages of N-OPA

- Complexity: The N-OPA-based decision-making process incorporates multiple matrices and priorities, which can introduce complexity.
- Knowledge Prerequisite: N-OPA relies on the availability of experts with knowledge and experience in the decision-making field. Obtaining such expertise can be challenging in some situations.
- Data Requirement: N-OPA needs a large amount of data to produce precise and accurate decisions. Using N-OPA in situations where data are scarce can be challenging.
- Specialised Software and Tools: The complexity of the method may require specialised software and tools, which may be impractical or inaccessible for all decision makers.
- The following justifications can be used to support the choice of N-OPA for problem solving in disaster mitigation measures, with special emphasis on flood-prone areas, and why it is superior to other MCDM techniques:
- Full Treatment of Ambiguity and Uncertainty: N-OPA fully considers these factors for a more realistic analysis, whilst other methods either struggle with them or neglect them entirely.
- No Demand for Exact Values, Instead Uses Ordered Rankings: Other techniques require that the exact numerical value of each criterion be provided. By contrast, N-OPA uses ordinal rankings, which evades the limitations of exact quantification.
- Handling of Numerous Criteria: Many decision-making techniques face challenges when multiple criteria are present. The MCDM technique N-OPA can handle this circumstance.
- Greater Sensitivity to Changes (Flexibility): N-OPA is extremely adaptable and versatile, which are essential in dynamic situations like disaster mitigation techniques for flood-prone areas.

# 3.5.3 Steps in N-OPA for evaluating main criteria and sub-criteria

-The appropriate panel of experts is defined, and the clear objectives of the study are identified. -The expert panel selects the suitable set of criteria and sub-criteria related to the study domain.

-The relative importance ratings given by the panel of experts to the criteria and sub-criteria are collected. As shown in Table 2, each expert must rank the criteria and sub-criteria in the form of triangular neutrosophic numbers (TNNs) (L, M, U; CD).

#### Table 2

Linguistic variables for determining the importance degree of criteria and sub-criteria.

Linguistic terms	Lower, median, and upper value of	Confirmation degree of expert
Elliguistic ter llis	number (L, M, U)	Opinion (CD)
Absolutely not important	$\langle (0, 0, 0) \rangle$	Absolutely not sure (0,1,1)
Not important	$\langle (0,0,1) \rangle$	Not sure (0.25 ,0.75, 0.75)
Slightly important	$\langle (1, 2, 3) \rangle$	Slightly sure (0.45, 0.60, 0.60)
Median important	$\langle (2,3,4) \rangle$	Median sure (0. 50 , 0. 50, 0. 50)
Important	$\langle (3,4,5) \rangle$	Sure (0.75, 0.20, 0.20)
Strongly important	<b>⟨(5, 6, 7)⟩</b>	Strongly sure (0.85, 0.15, 0.15)
Very strongly important	$\langle (6,7,8) \rangle$	Very strongly sure (0.90, 0.10, 0.10)
Absolutely important	$\langle (7, 8, 9) \rangle$	Absolutely sure (1.00, 0.00, 0.0)

In the presence of multiple experts, their individual perspectives will be combined using an aggregation method. The aggregator operator, which is symbolised as 'G', acts as a mapping function expressed as  $G: \psi n \rightarrow \psi$ . In other words, 'G' helps transition from multiple opinions to a single one. During this aggregation procedure, the focus is on deriving the average value of each importance degree of criteria and sub-criteria. This procedure is achieved by dividing the total importance of each attribute 'j' by the number 'k' of experts or decision makers. Here, 'n' represents the total count of the experts or decision makers involved in this process.

As shown in Table 2, each expert needs to rank the criteria and sub-criteria using TNNs, which are expressed as (L, M, U; CD). For instance, if the expert ranks the first criterion as superior, then it is assigned as 'absolutely important' linguistic variable. If the expert is extremely confident about the ranking of this first criterion, then the final assessment value will adopt the form of the TNN (7, 8, 9); 0.90, 0.10, 0.10. The first part of this representation (7, 8, 9) represents the lower, median and upper limits of the TNN. The second part (0.90, 0.10, 0.10) stands for the degree of confirmation, which incorporates the maximum degree of truth, minimal abstruseness and falsity levels associated with a triangular number.

After the opinions of experts on the relative importance of the criteria and sub-criteria are aggregated, a score function equation is used to determine the final ranking or priority of the consolidated values. For instance, let  $A^{\sim} = \langle (L, M, U); \mu, \gamma, \lambda \rangle$  be a TNN. Then, the score function is calculated from Eq.(1) as follows:

 $S(A) = 1/12 \ (L + 2M + U) * [2 + \mu - \gamma - \lambda]$ 

The subsequent score function generated from these criteria and sub-criteria is used to determine priority. If the score of C<sup>-1</sup> is higher than that of C<sup>-2</sup>, then C<sup>-1</sup> would have a higher priority than C<sup>-2</sup>. Similarly, if the score of C<sup>-1</sup> is less than that of C<sup>-2</sup>, then C<sup>-1</sup> would have a lower priority. If the scores of C<sup>-1</sup> and C<sup>-2</sup> are equal, then both of them hold the same priority.

After the criteria and sub-criteria are ranked based on their aggregated final rank, the traditional ordinal priority approach (OPA) is used. OPA is a decision-making process used to determine priorities, where the alternatives or options are ranked in order of preference or importance, but the difference between each rank may not be equal or easily quantifiable. In this approach, alternatives are compared not in terms of their exact values, but as being better, worse or equivalent to other options. This approach can be used when other sub-criteria or factors contribute to the ranking process, and the scores on them are not easily comparable. To use OPA, weights need to be assigned to each of the criteria or sub-criteria being evaluated. Then, the options need to be ranked according to their level of compliance with each criterion. The options are subsequently compared, and the one with the highest composite rank is considered the most suitable option. The three main steps of OPA are criteria and sub-criteria identification, weighting of the criteria and ranking of alternatives. This approach is often used in situations where decision makers have limited information or data about

(1)

the alternatives or where quantitative methods are inappropriate. The constructed model, according to Equations 2–9 introduced by Abdel-Basset *et al.* [81] is as follows:

-This explanation elucidates the ranking of sub-criteria A based on various 'c' sets of criteria values. The equation is formatted from Eq.(2) as follows:

 $A_{ca}^{1} \ge A_{ca}^{2} \ge \dots \ge A_{ca}^{r} \ge A_{ca}^{r+1} \ge \dots \ge A_{ca}^{m} \qquad \forall c, a$ <sup>(2)</sup>

-The dominance of the  $i^{th}$  sub-criteria over the  $l^{th}$  sub-criteria follows a similar structure, with the weights ( $W_{ca}$ ) replacing the sub-criteria:

 $W_{ca}^{1} \ge W_{ca}^{2} \ge \dots \ge W_{ca}^{r} \ge W_{ca}^{r+1} \ge \dots \ge W_{ca}^{n} \qquad \forall c, a$ (3)

-To demonstrate the significance of sub-criteria weights, both sides of the equation are multiplied by 'c' with rank 'r' as calculated in Eq.(4):

 $c(r(W_{ca}^{r}-W_{ca}^{r+1})) \geq 0,$ 

-To compute the weights of the sub-criteria, a model is proposed in which the reference for each criterion is maximised, subject to the total weights equalling 1 and each weight being positive as in Eq.(5).

$$\begin{aligned} & \text{Max} \{ c \big( r (W_{ca}^r - W_{ca}^{r+1}) \big), cm W_{ca}^m \} \\ & \text{Subject to:} \\ & \sum_{c=1}^n \sum_{a=1}^m W_{ca}^r = 1 \\ & W_{ca}^r \geq 0, \end{aligned} \tag{5}$$

-Given that the model is multi-objective and nonlinear and aims to maximise the minimisation objectives, it now reads as Eq.(6).

Max Min {
$$c(r(W_{ca}^{r} - W_{ca}^{r+1})), cmW_{ca}^{m}$$
}  
Subject to  
 $\sum_{c=1}^{n} \sum_{a=1}^{m} W_{ca}^{r} = 1$   
 $W_{ca}^{r} \ge 0,$ 
(6)

-To express the model linearly in Eq.(7):

Max Z

Subject to  

$$c(r(W_{ca}^{r} - W_{ca}^{r+1})) \ge Z$$

$$cmW_{ca}^{m} \ge Z$$

$$\sum_{c=1}^{n} \sum_{a=1}^{m} W_{ca}^{r} = 1$$

$$W_{ca}^{r} \ge 0$$
(7)

where  $Z = Min\{c(r(W_{ca}^{r} - W_{ca}^{r+1})), cmW_{ca}^{m}\}$  and it is unrestricted in sign.

-This model can be solved using various software solutions like LINGO or Excel. The weights of the sub-criteria and criteria are calculated as follows in Eq.(8) and Eq.(9):

$W_a = \sum_{c=1}^{n} W_{ca}^r$	∀ a,	(8)
$W_c = \sum_{r=1}^m W_{ca}^r$	∀ c,	(9)

-The final rank of the sub-criteria can be then obtained based on their global weight.

N-OPA is appropriate for this case study specifically due to the following reasons: **Uncertain Information:** Owing to various factors, such as weather conditions and drone battery life, uncertainty

(4)

could exist in the information regarding optimal locations for takeoff and landing. N-OPA can handle this issue effectively. **Multiple Experts**: Decisions on drone navigation could involve inputs from multiple experts. N-OPA aggregates the rankings given by various experts, which provides an overall judgment. **Variable Criteria**: Aspects, such as terrain, wind speed and proximity to populated areas, could all influence the optimal location decision. N-OPA can consider multiple criteria for decision making.

### 4. Case Study

Flooding on the Egyptian Mediterranean Coast has led to notable losses, which require a proactive approach to analyse vulnerable areas. The complexities of environmental patterns and the absence of real-time data, inherent uncertainty and the high risks associated with conventional surveillance increase this difficulty. In addition, finding locations for drone takeoff and landing presents challenges. UAVs or drones are crucial in disaster management, especially in hard-to-reach areas. The selection of suitable sites for drone operation influences operational efficiency, area coverage, data collection effectiveness and the safe use of drones, especially considering the unique geographic and demographic characteristics of Egypt. Effective disaster management depends on overcoming these obstacles and establishing a reliable method for the selection of drone locations. This study proposes a multifaceted approach to address these challenges within a comprehensive research framework aimed at providing a unified disaster management strategy.

The case study focuses on the use of IoT, GIS and drones for flood risk management in the Egyptian Mediterranean Coast. This region is prone to flooding due to its geographical location, with several towns and cities experiencing severe flooding during the winter months. This study aims to develop a framework for selecting optimal drone sites using MCDM to enhance flood management in the region. This study utilised historical data from the past decade on flood occurrences in the region to identify the most vulnerable areas. Furthermore, hydrological and meteorological data were collected from weather stations to predict flooding. Spatial data, such as satellite images, DEMs and land cover maps, were also used to map flood-prone areas. This study assumes that drone technology is available and feasible for deployment during flood events. In addition, IoT and GIS technologies are effective tools in flood monitoring and modelling. Lastly, the study assumes that the MCDM framework is suitable for the selection of optimal drone sites in flood management.

### 4.1 Determination of Flood-Prone Governorates

# 4.1.1 Curve Fit Forecast

In ArcGIS Pro 3.1, the curve fit forecast tool works by applying one of four equations or curves to the training dataset. Curve fit mathematical models are used to describe the relationship between variables in a dataset and predict future data points. These models use mathematical formulas adjusted to best fit the data and include [82]:

- Linear Regression: This model assumes a linear relationship between the dependent and the independent variable. It is represented by the equation y = mx + c, where m and c are the parameters, our model will try to learn to produce the best fit line.
- **Exponential Regression**: It is used when data show exponential growth or decay. The generic form of its equation is in Eq. (10)
- $y = ab^X$ 
  - **Parabolic Curve**: In a parabolic curve fit, the data are modelled to a quadratic function of the form as in Eq. (11)

$$y = ax^2 + bx + c \tag{11}$$

21

(10)

This model can capture rising and falling trends in the data that other simple models like linear fit may not successfully represent.

• **Gompertz Model**: It is also called Gompertz function. It is a type of mathematical model for a time series and is often used in forecasting scenarios, especially when the growth is expected to asymptote at a maximum level. Its equation is usually expressed in the form in Eq. (12).

 $y = ae^{-be^{-ct}}$ 

(12)

Where a, b and c are the parameters of the function fit by the data.

A single curve type can be selected manually to apply it to all the locations in the cube, or it can be auto-detected by the software, with each location in the cube using different equation or curve that is the best fit. In the curve fit forecast tool, the following parameters were selected: the analysis variables were historical hourly humidity and rainfall datasets from 2003 to 2023.

Humidity and rainfall are crucial prediction parameters when forecasting floods in a prone area due to the following reasons: **Direct Impact on Floods**: Rainfall is the most direct contributor to floods. Excessive rainfall, particularly in a short period, can overwhelm the ability of the soil to absorb it, which leads to runoff that fills rivers, streams, and other water bodies. This phenomenon often causes them to overflow their banks or the drainage capacities in urban areas. **Soil Saturation**: Humidity levels can provide information on the amount of moisture already present in the environment. High humidity levels might indicate recent rainfall, and the soil could already be saturated. Saturated or nearly saturated soil is less able to absorb additional rainfall, which increases the likelihood of flooding. **Predicting Rainfall:** Higher humidity levels often precede rainfall. Thus, monitoring humidity can help anticipate periods of heavy rains, which increases the accuracy of flood prediction models. **Climate Patterns**: Both parameters also influence longer-term climate patterns, which can have implications for flood risk. Extended periods of high humidity can lead to increased cloud formation and frequent rainfall, which raise the likelihood of flood events.

By monitoring changes in these parameters, emergency planners can improve the accuracy of their flood predictions, deploy resources more effectively and potentially save lives and property by warning residents in a timely manner. Other parameters included selecting number of time steps to forecast in this study; 63 or 5 years were selected. In the meantime, the selected curve type was auto-detected by the tool. Figure 5 shows the forecast results using the curve fit forecast.



Fig. 5. Prediction using curve fit forecast (left: rainfall variable, right: humidity variable)

#### 4.1.2 Exponential Smoothing

In ArcGIS Pro 3.1, the exponential smoothing tool functions by utilising past observations to forecast future points in the time series. The method estimates three components, namely, level, trend and seasonal, depending on the selected model type. Each time step influences the next time step in an exponentially decreasing manner. Users can specify the season span. In this study, a span of 4 months was used. Parameters, such as the input feature layer, time field, value field(s), prediction length and confidence level for prediction intervals, were defined following the previous method. Figure 6 shows the forecast results.



Fig. 6. Prediction using exponential smoothing (left: rainfall variable, right: humidity variable)

#### 4.1.3 Forest-based Forecast

In ArcGIS Pro 3.1, the 'Forest-based Classification and Regression' tool uses a machine learning method based on decision trees known as random forest. The tool creates a multitude of decision trees and combines their outcomes to produce either a classification or regression analysis. The sliding time window parameter guides the tool with the number of time steps to use as the training set. The model runs location by location across the space–time cube, and the tool can be used to select the appropriate time step window. Other parameters were defined similarly to the previous tools in running the tool. The results are shown in Figure 7.



Fig. 7. Prediction using forest-based forecast (left: rainfall variable, right: humidity variable)

#### 4.1.4 Inception Time

Although curve fitting, exponential smoothing and forest-based forecasting are excellent forecasting methods, they may fall short in handling certain types of complex data structures. InceptionTime, which is a type of CDNN, excels in this aspect. In ArcGIS Pro 3.1, before we can run the CDNN model, we firstly need to use another tool to prepare the datasets, which is called the train time series forecast model. It uses time series data from a space—time cube to build a deep learning-based time series forecasting model. Then, the outcomes with the previously selected variables can be used to run the forecast using the time series model based on the CDNN method InceptionTime for prediction. The results are shown in Figure 8.



Fig. 8. Prediction using InceptionTime (left: rainfall variable, right: humidity variable)

#### 4.1.5 Model Validation and Evaluation

Root mean square error (RMSE) is a frequently used measure of the differences between values predicted by a model and the values actually observed from the environment being modelled. In other words, it quantifies how closely the data align with the line of best fit [83].

**Forecast RMSE (FRMSE):** The FRMSE quantifies the difference between the values predicted by the model and the observed values during the training phase of the modelling process. Essentially, it assesses how accurately the model predicts sample data during training.

**Validation RMSE (VRMSE)**: The VRMSE represents the difference between the values predicted by the model and the observed values on a hold-out validation set. This validation set consists of samples that were not used during the training phase and were set aside for testing. VRMSE is particularly useful because it provides insights into how well the model is likely to perform on new, unseen data.

As shown in Table 3, a lower FRMSE suggests a better fit to the data because it means the predictions of the model closely match the observed data. However, RMSE is a scale-dependent measure, which means whether a given RMSE value is considered good, or poor depends on the context, or the variable being predicted. It is also sensitive to outlier values, which suggests that a single highly inaccurate prediction can largely impact the FRMSE [84].

A low VRMSE suggests that the predictions of the model closely align with the actual values in the validation dataset, which indicates good generalisability. Therefore, the model can accurately predict new, unseen data. Conversely, a high VRMSE suggests significant disparities between the predictions of the model and the actual values in the validation dataset. This condition may suggest overfitting, where the model is too closely fitted to the training data and performs poorly on new, unseen data [85].

RMSE and VRMSE for prediction models							
	Humidity	v Variable	Rainfall	Variable			
Forecast Approach	FRMSE	VRMSE	FRMSE	VRMSE			
Curve Fit Forecast	3.482	3.648	0.027	0.038			
Exponential Smoothing	3.501	3.68	0.027	0.038			
Forest-based Forecast	0.285	0.807	0.01	0.031			
InceptionTime	2.096	0.732	0.032	0.02			

 Table 3

 FRMSE and VRMSE for prediction models

#### 4.1.6 Forecast Evaluation by Location

In ArcGIS Pro 3.1, a tool called 'Evaluate Forecast by Location' is used to select the most accurate forecasting approach amongst multiple forecasting results for each location of a space-time cube. This tool enables the comparison of multiple tools in the Time Series Forecasting toolset using the same time series data and helps in selecting the best forecast for each location. Figure 9 shows the results obtained from the evaluation tool, which indicates that the forest-based forecast provides the best fit.



Fig. 9. Evaluation forecast by locations

#### 4.2 Application of N-OPA

After comparing predictions generated using different approaches, the governorate most prone to floods in the next 5 years is Port Said. A combination of the MCDM method N-OPA and GIS processing was employed to identify the most suitable locations for drone takeoff and landing in Port Said for flood monitoring.

N-OPA was used to develop the weights for criteria and sub-criteria. The main criteria were assessed by all experts using linguistic variables, as previously presented in Table 2. The evaluation of the primary criteria by all experts is exhibited in Table 4. Tables A.1–A.5 (Appendix A) present the evaluation of sub-criteria related to each main criterion by all experts.

#### Table 4

Evaluation values of criteria according to each expert

Е	(C1)	(C2)	(C3)	(C4)	(C5)	Е	(C1)	(C2)	(C3)	(C4)	(C5)
1	((6,7,8);	((5, 6, 7);	((2, 3, 4); (	((3, 4, 5);	((7,8,9);	4	((5,6,7);	((6,7,8);	((3, 4,5); (	((5, 6, 7); (	((6,7,8);
2	((6,7,8);	((2, 3, 4); (	((2, 3, 4); (	((2,3,4);	((7,8,9);	5	((5,6,7);	((3, 4,5);	((5, 6, 7);	((5, 6, 7); (	((6,7,8);
3	((7,8,9);	((6,7,8);	((6,7,8);	((3, 4, 5);	((7,8,9);						

We utilised the previously mentioned Equation (1) to determine the final rank of the aggregated value after the averaging of expert opinions regarding the significance of the criteria, as shown in Table 5.

#### Table 5

Neutrosophic computations of criteria for the ranking process

Computations			Criteria		
	(C1)	(C2)	(C3)	(C4)	(C5)
Sum of importance	(29,34,39;	(22,27,32;	(18,23,28;	(18,23,28;	(33,40,43;
degree for criteria	0.45,0.60,0.60)	0.25,0.75,0.75)	0.25,0.75,0.75)	0.45,0.60,0.60)	0.75,0.20,0.20)
Score value of average	2.83	1.35	1.15	1.91	6.11
importance degree					
Final rank of criteria	2	4	5	3	1

The N-OPA model was constructed, and the total weight of the main criteria was calculated using LINGO software, as shown in Figure 10. The relative importance of sub-criteria according to each criterion is shown in Table 6.



Fig. 10. Weights of the main criteria

Main criteria	Weights of main objectives	Sub- criteria	Local weight of sub-criteria	Rank	Global weight	Rank global weights
<b>C1</b>	0.252	A11	0.359	2	0.090	1
		A12	0.641	1	0.162	10
C2	0.093	A21	0.633	1	0.059	6
		A22	0.367	2	0.349	11
С3	0.072	A31	0.748	1	0.054	5
		A32	0.015	3	0.001	2
		A33	0.252	2	0.018	3
C4	0.148	A41	0.578	1	0.086	9
		A42	0.422	2	0.062	7
C5	0.434	A51	0.805	1	0.034	4
		A52	0.195	2	0.085	8

#### Table 6



### 4.3 GIS Processing

GIS is a system used for capturing, managing, analysing and presenting geospatial data. GIS is important because it allows researchers and disaster management professionals to visualise geospatial data, analyse patterns and relationships and ultimately make informed decisions. GIS helps integrate data from multiple sources and present them in a unified and understandable format. In the context of disaster risk management, GIS can help identify high-risk areas, plan evacuation routes and assess the impact of disasters. Overall, GIS is an essential tool for effective disaster risk management and can help save lives and minimise damage. In this study, GIS plays a crucial role in providing a framework for disaster risk management. Specifically,

- 1 The region most prone to flooding is predicted by analysing historical data and identifying the governorate with the highest vulnerability to flooding using GIS and four forecasting models.
  - GIS is used for visualising the collected historical dataset involved in the prediction phase.
  - Pre-processing of the dataset, such as how missing values are handled within the dataset, is conducted via GIS tools, specifically ArcGIS Pro 3.1.
  - The datasets collected are then converted from variable locations into a point shape file using conversion tools in ArcGIS Pro 3.1.
  - Space-time cubes are created from the point shape files using ArcGIS Pro 3.1 tools.
  - Four forecasting models within the ArcGIS Pro 3.1 software are then applied to the dataset.
  - Model validation and evaluation are conducted using tools within ArcGIS.
- 2 The best locations for drone takeoff and landing are selected using GIS with MCDM to maximise the efficiency and effectiveness of data collection.
  - GIS is used in the second phase for the construction and standardisation of sub-criteria map layers using ArcGIS Pro 3.1 tools.
  - Sub-criteria map layers are created, and total weights are integrated using the raster calculator tool in ArcGIS Pro 3.1.
  - The final weighted layers are then used as inputs to the ArcGIS Pro 3.1 suitability modeler to locate the most suitable locations for drone takeoff and landing.
  - A total of 10 locations are selected to be suggested as suitable locations by the modeler based on the inputs, with a total area of 120 m<sup>2</sup> for each location, as shown in Figure 11.



Fig. 11. Suitable locations for drone takeoff and landing in Port Said, Egypt

IoT refers to the network of physical objects or 'things' that are embedded with sensors, software and other technologies that enable them to connect and exchange data over the Internet. In the context of disaster risk management, IoT can be used to gather real-time data before, during and after a disaster to inform decision-making and response efforts. Using IoT requires prior planning for more efficiency. In this study, IoT is suggested to be used in conjunction with GIS to enhance disaster risk management. The following steps are conducted to facilitate the actual implementation of the IoT framework:

- Drones equipped with sensors and data-transmission capabilities are used to gather realtime data on the impact of the disaster before, during and after it occurs.
- The optimal locations for drone takeoff and landing are selected using GIS with MCDM to maximise the efficiency and effectiveness of data collection. This study uses MCDM to select the optimal locations for drone takeoff and landing. This method is a decisionmaking technique that involves evaluating and prioritising multiple criteria to choose the best option. A neutrosophic-based MCDM method called N-OPA is used to handle uncertainty and evaluate criteria and sub-criteria in a neutrosophic environment.
- The drones are implemented at the selected locations, and the data collected by drones are analysed to inform decision-making and response efforts.

One limitation of the case study is that it focuses only on flooding in the Egyptian Mediterranean Coast, which may not be representative of all regions or types of disasters. The study also relies on data from open sources rather than commercial data with higher resolution, which may affect the accuracy of the forecasting models and drone site selection. The panel of experts involved in the MCDM process may also introduce bias in the selection of criteria and sub-criteria. Future research directions can expand the case study to different regions and types of disasters to test the generalisability of the framework. Real-time data can also be used to improve the accuracy of the predictions and site selection process. Lastly, exploring the ethical considerations and implications of using drones for disaster response and management can be a valuable avenue for future research.

#### 5. Conclusions

This study aims to achieve two primary objectives. Firstly, it aims to predict the region most prone to flooding using GIS in conjunction with forecasting models. Secondly, it aims to optimise the selection of drone takeoff and landing locations based on real-time IoT data. In this study, GIS with MCDM is used to harness the full potential of IoT real-time data obtained through drones. The research methodology employs a two-phase framework. Phase 1 focuses on predicting the governorate most prone to flooding. In this phase, GIS and four forecasting models—three traditional models and one based on CDNN—are used to analyse historical data and identify the governorate with the highest vulnerability to flooding. Phase 2 involves the selection of optimal locations for drone takeoff and landing using GIS with MCDM. This phase involves building an MCDM model by identifying suitable criteria and then collecting and processing the necessary data to implement the model. The criteria used in the MCDM model need to be prioritised by assigning weights to find the best locations. A neutrosophic-based MCDM method called N-OPA is used for weight assignment. The use of N-OPA, which is an extended form of the original OPA, handles decision-making uncertainties in a neutrosophic environment. GIS tools are then applied to find the total weights of all locations in the selected governorate. The top-ranked locations are identified as the most appropriate for drone takeoff and landing based on the weights. A case study from the Egyptian Mediterranean Coast is used to validate the effectiveness and applicability of the framework in disaster management strategies. Eight different governorates along the Egyptian Mediterranean Coast are selected to assess their vulnerability to flooding in the first phase of the framework. Predictions are made using four forecasting models, namely, curve fit forecast, exponential smoothing, forest-based forecast and InceptionTime. These predictions are based on historical hourly humidity and rainfall datasets from 2003 to 2023, which are acquired from the website for accessing data from NASA. The evaluate forecast by location tool in ArcGIS software is used to select the most accurate prediction model. The results indicate that the forest-based forecast provides the best fit, and the Port Said governorate is predicted as the most vulnerable to flooding. The second phase of the case study focuses on building an MCDM model to select the best sites across Port Said governorate for drone takeoff and landing. A panel of five experts with relative experience and knowledge assists in selecting the five main criteria and eleven sub-criteria for evaluation. Spatial data are collected from secondary sources, including NASA for DEM data, NARSS for road and coastline data, CAPMAS for airport and population data and the Breezometer website for air quality index information. Additional data, such as slope, restricted areas and land use, are generated using various analytic tools found in ArcGIS Pro 3.1 Pro software. The MCDM method N-OPA is then employed to evaluate the criteria and sub-criteria. Based on the weights assigned and using ArcGIS tools, locations in Port Said are ranked, and the top 10 suitable sites for takeoff and landing are recommended. Key findings from this research include the successful prediction of the most vulnerable governorate to flooding and the selection of top-ranked locations for drone takeoff and landing. The practical implications of this research suggest the use of similar frameworks to collect and analyse real-time data, which enables timely and effective decision making in response to disasters. Future research directions may explore the use of different IoT sensors and the application of the framework to different types of disasters. Furthermore, the actual implementation of the drones and the analysis of collected real-time data need to be considered. Although various predictive models are applied, our research is limited by the quantity and quality of available data, which potentially affects the accuracy of our results. Future research should expand on these findings by possibly integrating more complex models to address the unique challenges and dynamics of modern disaster management. Moreover, the case study focuses only on one type of disaster in one region and does not address the ethical considerations of using drones for disaster response and management. Overall, this study provides a comprehensive, rigorous and technologically advanced framework for enhancing disaster management strategies and can serve as a foundation for further research and practical applications in this field.

**Author Contributions:** Conceptualization, N.M.A., K.A.E., M.A.B. and S.A.S.; methodology, N.M.A., K.A.E. S.A.S., and M.A.B.; software, S.A.S.; validation, M.A.B., S.A.S. and A.G.; formal analysis, N.M.A.; investigation, M.A.B. and A.G.; resources, S.A.S.; data curation, N.M.A.; writing—original draft preparation, S.A.S.; writing—review and editing, N.M.A., K.A.E., S.A.S., A.G. and M.A.B.; visualization, M.A.B.; supervision, M.A.B.; project administration, M.A.B. All authors have read and agreed to the published version of the manuscript.

### Funding

No funding was received for this study.

### Data availability statement

The datasets generated during and/or analysed during the current study are not publicly available due to the privacy-preserving nature of the data. However, they can be obtained from the corresponding author upon reasonable request.

# **Conflict of interest**

The authors declare no conflicts of interest in this research.

## **Ethical approval**

This article does not involve any studies with human participants or animals performed by any of the authors.

### Appendix A

### Table A.1

Evaluation values of legal sub-criteria according to each expert

	0	0			
Exper ts	A11	A12	Exper ts	A11	A12
1	((5, 6, 7; 0.50, 0.50, 0.5)	((7, 8, 9); ((1.00, 0.00, 0.	4	((6, 7, 8; 0.85, 0.15, 0.1!	((7, 8, 9; (1.00, 0.00, 0.0
2	((6,7,8; 0.85 ,0.15,0.15)	((5, 6, 7; 0.50, 0.50, 0.50)	5	((5, 6, 7; 0.50, 0.50, 0.5)	((6, 7, 8; 0.85, 0.15, 0.1)
3	((6, 7, 8; 0.85, 0.15, 0.1)	((6,7,8); (1.00,0.0,0.0)			

### Table A.2

Evaluation values of accessibility sub-criteria according to each expert

Exper ts	A21	A22	Exper ts	A21	A22
1	((7, 8, 9; 1.00, 0.00, 0.0)	((5, 6, 7; 0.85, 0.15, 0.15)	4	((6, 7, 8; 0.75, 0.20, 0.20)	((3, 4,5; 0.50, 0.50, 0.50
2	((6,7,8); (1.00,0.0,0.0	((5, 6, 7; 0.85, 0.15, 0.15)	5	((3, 4,5; 0.50 ,0.50, 0.50	((5, 6, 7; 0.85, 0.15, 0.15)
3	((5, 6, 7; 0.50, 0.50, 0.50)	((6, 7, 8; 0.75, 0.20, 0.20)			

#### Table A.3

Evaluation values of topography sub-criteria according to each ex	xpert
---	-------

Exp	٨21	٨22	A33	Exp	٨21	٨22	A33
erts	ASI	A32		erts	ASI	A32	
1	((7, 8, 9; 1.00, 0.	((3, 4, 5; 0.50, 0.	((6, 7, 8; 0.75, 0.	4	((7, 8, 9; 0.85, 0.	((5, 6, 7; 0.85, 0.	((6, 7, 8; 0.75, 0.
2	((6, 7, 8; 0.75, 0.	((5, 6, 7; 0.85, 0.	((3, 4,5; 0.75,0.	5	((6, 7, 8; 0.75, 0.	((3, 4,5; 0.50, 0.	((5, 6, 7; 0.75, 0
3	((6, 7, 8; 0.85 ,0.	((5, 6, 7; 0.50, 0.	((7, 8, 9; 0.85, 0.				

#### Table A.4

Evaluation values of safety sub-criteria according to each expert

Expe rts	A41	A42	Experts	A41	A42
1	((5, 6, 7; 0.85 ,0.15,0	((7, 8, 9; 0.90, 0.10, 0	4	((7, 8, 9; 0.85, 0.15, 0)	((3, 4,5; 0.50, 0.50, 0
2	((3, 4,5; 0.75 ,0.20, 0	((6, 7, 8; 0.90, 0.10, 0	5	((6,7,8;0.90,0.10,0	((5, 6, 7; 0.85, 0.15, 0)
3	((7, 8, 9; 0.85, 0.15, 0)	((5, 6, 7; 0.50, 0.50, 0	((6, 7, 8; 0.85, 0.15, 0		

#### Table A.5

Evaluation values of environment sub-criteria according to each expert

Exper	۸51	۸52	Exper	۸51	۸52
ts	ASI	AJZ	ts	ASI	ASZ
1	((7, 8, 9; 1.00, 0.00, 0.0)	((5, 6, 7; 0.85 ,0.15,0.15	4	((7, 8, 9; 0.85 ,0.15,0.15	((6, 7, 8; 0.75, 0.20, 0.20)
2	((6, 7, 8; 0.90 ,0.10, 0.1(	((5, 6, 7; 0.85 ,0.15,0.15	5	((7, 8, 9; 0.90 ,0.10, 0.1(	((6, 7, 8; 0.90, 0.10, 0.10)
3	((6, 7, 8; 0.75, 0.20, 0.20)	((5, 6, 7; 0.50, 0.50, 0.50)			

#### References

- [1] Torani, S., Majd, P., Maroufi, S., Dowlati, M., & Sheikhi, R. (2019). The importance of education on disasters and emergencies: A review article. *Journal of Education and Health Promotion*, 8(1), 85. <u>https://doi.org/10.4103/jehp.jehp 262\_18</u>
- [2] Zuccaro, G., Leone, M. F., & Martucci, C. (2020). Future research and innovation priorities in the field of natural hazards, disaster risk reduction, disaster risk management and climate change adaptation: a shared vision from the ESPREssO project. International Journal of Disaster Risk Reduction, 51, 101783. https://doi.org/10.1016/j.ijdrr.2020.101783
- [3] Kim, Y., & Kim, M. Y. (2022). Factors affecting household disaster preparedness in South Korea. *PLOS ONE*, *17*(10), e0275540. <u>https://doi.org/10.1371/journal.pone.0275540</u>
- [4] Bachri, S., Sumarmi, Irawan, L. Y., Utaya, S., Wirawan, R., Nurdiansyah, F. D., Nurjanah, A. E., Tyas, L. W. N., Adillah, A. A., & Setia, D. (2022). FOSS (Free Open Source Software) Integration to Implement WebGIS-Based Information System of Kelud Volcano. *IOP Conference Series: Earth and Environmental Science*, 1066(1), 012010. <u>https://doi.org/10.1088/1755-1315/1066/1/012010</u>
- [5] Rehman, J., Sohaib, O., Asif, M., & Pradhan, B. (2019). Applying systems thinking to flood disaster management for a sustainable development. *International Journal of Disaster Risk Reduction*, 36, 101101. <u>https://doi.org/10.1016/j.ijdrr.2019.101101</u>
- [6] Rana, I. A. (2020). Disaster and climate change resilience: A bibliometric analysis. International Journal of Disaster Risk Reduction, 50, 101839. https://doi.org/10.1016/j.ijdrr.2020.101839
- [7] Rusk, J., Maharjan, A., Tiwari, P., Chen, T.-H. K., Shneiderman, S., Turin, M., & Seto, K. C. (2022). Multi-hazard susceptibility and exposure assessment of the Hindu Kush Himalaya. *Science of The Total Environment*, 804, 150039. <u>https://doi.org/10.1016/j.scitotenv.2021.150039</u>
- [8] Wang, Q., & Wang, H. (2022). Spatiotemporal dynamics and evolution relationships between land-use/land cover change and landscape pattern in response to rapid urban sprawl process: A case study in Wuhan, China. *Ecological Engineering*, 182, 106716. https://doi.org/10.1016/j.ecoleng.2022.106716
- [9] Molotoks, A., Smith, P., & Dawson, T. P. (2021). Impacts of land use, population, and climate change on global food security. *Food and Energy Security*, *10*(1), e261. https://doi.org/10.1002/fes3.261

- [10] Khan, W. Z., Rehman, M. H., Zangoti, H. M., Afzal, M. K., Armi, N., & Salah, K. (2020). Industrial internet of things: Recent advances, enabling technologies and open challenges. *Computers & Electrical Engineering*, 81, 106522. <u>https://doi.org/10.1016/j.compeleceng.2019.106522</u>
- [11] van Ginkel, M., & Biradar, C. (2021). Drought Early Warning in Agri-Food Systems. Climate, 9(9), 134. <u>https://doi.org/10.3390/cli9090134</u>
- [12] Yar, H., Hussain, T., Khan, Z. A., Koundal, D., Lee, M. Y., & Baik, S. W. (2021). Vision Sensor-Based Real-Time Fire Detection in Resource-Constrained IoT Environments. *Computational Intelligence and Neuroscience*, 2021, 1– 15. <u>https://doi.org/10.1155/2021/5195508</u>
- [13] Shah, S. A., Seker, D. Z., Hameed, S., & Draheim, D. (2019). The Rising Role of Big Data Analytics and IoT in Disaster Management: Recent Advances, Taxonomy and Prospects. *IEEE Access*, 7, 54595–54614. https://doi.org/10.1109/ACCESS.2019.2913340
- [14] Fedele, R., & Merenda, M. (2020). An IoT System for Social Distancing and Emergency Management in Smart Cities Using Multi-Sensor Data. Algorithms, 13(10), 254. https://doi.org/10.3390/a13100254
- [15] Ramesh, A., Rajkumar, S., & Jenila Livingston, L. M. (2020). Disaster Management in Smart Cities using IoT and Big Data. Journal of Physics: Conference Series, 1716(1), 012060. <u>https://doi.org/10.1088/1742-6596/1716/1/012060</u>
- [16] Gohari, A., Ahmad, A. Bin, Rahim, R. B. A., Supa'at, A. S. M., Abd Razak, S., & Gismalla, M. S. M. (2022). Involvement of Surveillance Drones in Smart Cities: A Systematic Review. *IEEE Access*, 10, 56611–56628. https://doi.org/10.1109/ACCESS.2022.3177904
- [17] Dhinakaran, D., Udhaya Sankar, S. M., Latha, B. C., Anns, A. E. J., & Sri, V. K. (2023). Dam Management and Disaster Monitoring System using IoT. 2023 International Conference on Sustainable Computing and Data Communication Systems (ICSCDS), 1197–1201. https://doi.org/10.1109/ICSCDS56580.2023.10105132
- [18] Han, S., Huang, H., Luo, Z., & Foropon, C. (2019). Harnessing the power of crowdsourcing and Internet of Things in disaster response. Annals of Operations Research, 283(1–2), 1175–1190. https://doi.org/10.1007/s10479-018-2884-1
- [19] Pradhan, M. (2019). Interoperability for Disaster Relief Operations in Smart City Environments. 2019 IEEE 5th World Forum on Internet of Things (WF-IoT), 711–714. https://doi.org/10.1109/WF-IoT.2019.8767169
- [20] Fawzy, D., Moussa, S. M., & Badr, N. L. (2022). The Internet of Things and Architectures of Big Data Analytics: Challenges of Intersection at Different Domains. *IEEE Access*, 10, 4969–4992. https://doi.org/10.1109/ACCESS.2022.3140409
- [21] Zhang, F., Qiao, Q., Wang, J., & Liu, P. (2022). Data-driven AI emergency planning in process industry. *Journal of Loss Prevention in the Process Industries*, *76*, 104740. https://doi.org/10.1016/j.jlp.2022.104740
- [22] Jiang, H. (2019). Mobile Fire Evacuation System for Large Public Buildings Based on Artificial Intelligence and IoT. IEEE Access, 7, 64101–64109. https://doi.org/10.1109/ACCESS.2019.2915241
- [23] Guo, K., & Zhang, L. (2022a). Adaptive multi-objective optimization for emergency evacuation at metro stations. *Reliability Engineering & System Safety*, 219, 108210. https://doi.org/10.1016/j.ress.2021.108210
- [24] Rabiei, P., Arias-Aranda, D., & Stantchev, V. (2023). Introducing a novel multi-objective optimization model for volunteer assignment in the post-disaster phase: Combining fuzzy inference systems with NSGA-II and NRGA. *Expert Systems with Applications*, 226, 120142. https://doi.org/10.1016/j.eswa.2023.120142
- [25] Mahmoodi, A., Jasemi Zergani, M., Hashemi, L., & Millar, R. (2022). Analysis of optimized response time in a new disaster management model by applying metaheuristic and exact methods. *Smart and Resilient Transportation*, 4(1), 22–42. https://doi.org/10.1108/SRT-01-2021-0002
- [26] Ogunjinmi, P. D., Park, S. S., Kim, B., & Lee, D. E. (2022). Rapid Post-Earthquake Structural Damage Assessment Using Convolutional Neural Networks and Transfer Learning. *Sensors*, 22(9), 3471. <u>https://doi.org/10.3390/s22093471</u>
- Youssef, A. M., Pradhan, B., Dikshit, A., & Mahdi, A. M. (2022). Comparative study of convolutional neural network (CNN) and support vector machine (SVM) for flood susceptibility mapping: a case study at Ras Gharib, Red Sea, Egypt. *Geocarto International*, 37(26), 11088–11115. https://doi.org/10.1080/10106049.2022.2046866
- [28] Kaur, S., Gupta, S., Singh, S., & Arora, T. (2022). A Review on Natural Disaster Detection in Social Media and Satellite Imagery Using Machine Learning and Deep Learning. *International Journal of Image and Graphics*, 22(05), 2250040. https://doi.org/10.1142/S0219467822500401

- [29] Aliahmadi, A., Nozari, H., Ghahremani-Nahr, J., & Szmelter-Jarosz, A. (2022). Evaluation of key impression of resilient supply chain based on artificial intelligence of things (AIoT). *Journal of Fuzzy Extension and Applications*, 3(3), 201–211. https://doi.org/10.22105/jfea.2022.345008.1221
- [30] ortino, G., Savaglio, C., Spezzano, G., & Zhou, M. (2021). Internet of Things as System of Systems: A Review of Methodologies, Frameworks, Platforms, and Tools. *IEEE Transactions on Systems, Man, and Cybernetics: Systems*, 51(1), 223–236. https://doi.org/10.1109/TSMC.2020.3042898
- [31] De Donno, M., Tange, K., & Dragoni, N. (2019). Foundations and Evolution of Modern Computing Paradigms: Cloud, IoT, Edge, and Fog. IEEE Access, 7, 150936–150948. https://doi.org/10.1109/ACCESS.2019.2947652
- [32] Khan, R., Shabaz, M., Hussain, S., Ahmad, F., & Mishra, P. (2022). Early flood detection and rescue using bioinformatic devices, internet of things (IOT) and Android application. *World Journal of Engineering*, 19(2), 204–215. <u>https://doi.org/10.1108/WIE-05-2021-0269</u>
- [33] Anon, K., Ranjith, V., & Varalakshmi, K. (2021). IOT INTEGRATED FOREST FIRE DETECTION AND PREDICTION USING NODE MCU. International Journal of Electronics Engineering and Applications, IX(I), 28. <u>https://doi.org/10.30696/IJEEA.IX.I.2021.28-35</u>
- [34] Khan, A., Gupta, S., & Gupta, S. K. (2020). Multi-hazard disaster studies: Monitoring, detection, recovery, and management, based on emerging technologies and optimal techniques. *International Journal of Disaster Risk Reduction*, 47, 101642. https://doi.org/10.1016/j.ijdrr.2020.101642
- [35] Esposito, M., Palma, L., Belli, A., Sabbatini, L., & Pierleoni, P. (2022). Recent Advances in Internet of Things Solutions for Early Warning Systems: A Review. Sensors, 22(6), 2124. <u>https://doi.org/10.3390/s22062124</u>
- [36] Hajjaji, Y., Boulila, W., Farah, I. R., Romdhani, I., & Hussain, A. (2021). Big data and IoT-based applications in smart environments: A systematic review. *Computer Science Review*, 39, 100318. <u>https://doi.org/10.1016/j.cosrev.2020.100318</u>
- [37] Lee, S., Gandla, S., Naqi, M., Jung, U., Youn, H., Pyun, D., Rhee, Y., Kang, S., Kwon, H.-J., Kim, H., Lee, M. G., & Kim, S. (2020). All-Day Mobile Healthcare Monitoring System Based on Heterogeneous Stretchable Sensors for Medical Emergency. *IEEE Transactions on Industrial Electronics*, 67(10), 8808–8816. <u>https://doi.org/10.1109/TIE.2019.2950842</u>
- [38] Solpico, D., Tan, M. I., Manalansan, E. J., Zagala, F. A., Leceta, J. A., Lanuza, D. F., Bernal, J., Ramos, R. D., Villareal, R. J., Cruz, X. M., dela Cruz, J. A., Lagazo, D. J., Honrado, J. L., Abrajano, G., Libatique, N. J., & Tangonan, G. (2019). Application of the V-HUB Standard using LoRa Beacons, Mobile Cloud, UAVs, and DTN for Disaster-Resilient Communications. 2019 IEEE Global Humanitarian Technology Conference (GHTC), 1–8. https://doi.org/10.1109/GHTC46095.2019.9033139
- [39] Cui, F. (2020). Deployment and integration of smart sensors with IoT devices detecting fire disasters in huge forest environment. *Computer Communications, 150,* 818–827. <u>https://doi.org/10.1016/j.comcom.2019.11.051</u>
- [40] Pradhan, M. (2021). Federation Based on MQTT for Urban Humanitarian Assistance and Disaster Recovery Operations. *IEEE Communications Magazine*, 59(2), 43–49. https://doi.org/10.1109/MCOM.001.2000937
- [41] Siddiqi, Yu, & Joung. (2019). 5G Ultra-Reliable Low-Latency Communication Implementation Challenges and Operational Issues with IoT Devices. *Electronics*, 8(9), 981. https://doi.org/10.3390/electronics8090981
- [42] Mao, W., Zhao, Z., Chang, Z., Min, G., & Gao, W. (2021). Energy-Efficient Industrial Internet of Things: Overview and Open Issues. *IEEE Transactions on Industrial Informatics*, 17(11), 7225–7237. https://doi.org/10.1109/TII.2021.3067026
- [43] Li, W., Batty, M., & Goodchild, M. F. (2020). Real-time GIS for smart cities. International Journal of Geographical Information Science, 34(2), 311–324. https://doi.org/10.1080/13658816.2019.1673397
- [44] Breunig, M., Bradley, P. E., Jahn, M., Kuper, P., Mazroob, N., Rösch, N., Al-Doori, M., Stefanakis, E., & Jadidi, M. (2020). Geospatial Data Management Research: Progress and Future Directions. *ISPRS International Journal of Geo-Information*, 9(2), 95. https://doi.org/10.3390/ijgi9020095
- [45] Nabil M. AbdelAziz, & Safa Al-Saeed. (2023). Mitigating Landslide Hazards in Qena Governorate of Egypt: A GISbased Neutrosophic PAPRIKA Approach. *Neutrosophic Systems with Applications*, 7(SE-Articles), 13–35. <u>https://doi.org/10.61356/j.nswa.2023.37</u>

- [46] Mukherjee, M., Abhinay, K., Rahman, M. M., Yangdhen, S., Sen, S., Adhikari, B. R., Nianthi, R., Sachdev, S., & Shaw, R. (2023). Extent and evaluation of critical infrastructure, the status of resilience and its future dimensions in South Asia. *Progress in Disaster Science*, 17, 100275. <u>https://doi.org/10.1016/j.pdisas.2023.100275</u>
- [47] Hawchar, L., Naughton, O., Nolan, P., Stewart, M. G., & Ryan, P. C. (2020). A GIS-based framework for high-level climate change risk assessment of critical infrastructure. *Climate Risk Management*, 29, 100235. <u>https://doi.org/10.1016/j.crm.2020.100235</u>
- [48] Saini, K., Kalra, S., & Sood, S. K. (2023). Fog-inspired framework for emergency rescue operations in post-disaster scenario. *The Journal of Supercomputing*. https://doi.org/10.1007/s11227-023-05475-x
- [49] -Köhle, M., Cristofari, G., Wenk, M., & Fuchs, S. (2019). The importance of indicator weights for vulnerability indices and implications for decision making in disaster management. *International Journal of Disaster Risk Reduction*, 36, 101103. https://doi.org/10.1016/j.ijdrr.2019.101103
- [50] Safa Al-Saeed, & Nabil M. AbdelAziz. (2023). Integrated Neutrosophic Best-Worst Method for Comprehensive Analysis and Ranking of Flood Risks: A Case Study Approach from Aswan, Egypt. *Neutrosophic Systems with Applications*, 5(SE-Articles), 10–26. <u>https://doi.org/10.61356/j.nswa.2023.24</u>
- [51] Bejani, M. M., & Ghatee, M. (2021). A systematic review on overfitting control in shallow and deep neural networks. *Artificial Intelligence Review*, 54(8), 6391–6438. https://doi.org/10.1007/s10462-021-09975-1
- [52] Devaraj, J., Ganesan, S., Elavarasan, R., & Subramaniam, U. (2021). A Novel Deep Learning Based Model for Tropical Intensity Estimation and Post-Disaster Management of Hurricanes. *Applied Sciences*, 11(9), 4129. <u>https://doi.org/10.3390/app11094129</u>
- [53] Kyrkou, C., & Theocharides, T. (2019). Deep-Learning-Based Aerial Image Classification for Emergency Response Applications Using Unmanned Aerial Vehicles. 2019 IEEE/CVF Conference on Computer Vision and Pattern Recognition Workshops (CVPRW), 2019-June, 517–525. https://doi.org/10.1109/CVPRW.2019.00077
- [54] Kyrkou, C., & Theocharides, T. (2020). EmergencyNet: Efficient Aerial Image Classification for Drone-Based Emergency Monitoring Using Atrous Convolutional Feature Fusion. *IEEE Journal of Selected Topics in Applied Earth Observations and Remote Sensing*, 13, 1687–1699. <u>https://doi.org/10.1109/ISTARS.2020.2969809</u>
- [55] Munawar, H. S., Ullah, F., Qayyum, S., Khan, S. I., & Mojtahedi, M. (2021). UAVs in Disaster Management: Application of Integrated Aerial Imagery and Convolutional Neural Network for Flood Detection. *Sustainability*, 13(14), 7547. https://doi.org/10.3390/su13147547
- [56] Anbarasan, M., Muthu, B., Sivaparthipan, C. B., Sundarasekar, R., Kadry, S., Krishnamoorthy, S., Samuel R., D. J., & Dasel, A. A. (2020). Detection of flood disaster system based on IoT, big data and convolutional deep neural network. *Computer Communications*, *150*, 150–157. <u>https://doi.org/10.1016/j.comcom.2019.11.022</u>
- [57] Patil, P. G., Elluru, V., & Shivashankar, S. (2023). A new approach to MCDM problems by fuzzy binary soft sets. Journal of Fuzzy Extension and Applications. <u>https://doi.org/10.22105/jfea.2023.390059.1257</u>
- [58] Ibrahim M. Hezam. (2023). An Intelligent Decision Support Model for Optimal Selection of Machine Tool under Uncertainty: Recent Trends. *Neutrosophic Systems with Applications*, 3(SE-Articles), 35–44. https://doi.org/10.61356/j.nswa.2023.12
- [59] Sarma, D., Das, A., Dutta, P., & Bera, U. K. (2022). A Cost Minimization Resource Allocation Model for Disaster Relief Operations With an Information Crowdsourcing-Based MCDM Approach. *IEEE Transactions on Engineering Management*, 69(5), 2454–2474. https://doi.org/10.1109/TEM.2020.3015775
- [60] Gamal, A., & Mohamed, M. (2023). A Hybrid MCDM Approach for Industrial Robots Selection for the Automotive Industry. *Neutrosophic Systems with Applications*, 4(SE-Articles), 1–11. https://doi.org/10.61356/j.nswa.2023.13
- [61] Waleed Tawfiq Al-Nami. (2023). Ranking and Analysis the Strategies of Crowd Management to Reduce the Risks of Crushes and Stampedes in Crowded Environments and Ensure the Safety of Passengers. *Neutrosophic Systems with Applications*, 8(SE-Articles), 61–78. https://doi.org/10.61356/j.nswa.2023.50
- [62] Ortíz-Barrios, M., Jaramillo-Rueda, N., Gul, M., Yucesan, M., Jiménez-Delgado, G., & Alfaro-Saíz, J.-J. (2023). A Fuzzy Hybrid MCDM Approach for Assessing the Emergency Department Performance during the COVID-19 Outbreak. International Journal of Environmental Research and Public Health, 20(5), 4591. https://doi.org/10.3390/ijerph20054591

- [63] Guo, K., & Zhang, L. (2022b). Multi-objective optimization for improved project management: Current status and future directions. *Automation in Construction*, *139*, 104256. https://doi.org/10.1016/j.autcon.2022.104256
- [64] Ding, Z., Zhao, Z., Liu, D., & Cao, Y. (2021). Multi-objective scheduling of relief logistics based on swarm intelligence algorithms and spatio-temporal traffic flow. *Journal of Safety Science and Resilience*, 2(4), 222–229.

https://doi.org/10.1016/j.jnlssr.2021.07.003

- [65] Ershadi, M. M., & Shemirani, H. S. (2022). A multi-objective optimization model for logistic planning in the crisis response phase. *Journal of Humanitarian Logistics and Supply Chain Management*, *12*(1), 30–53. https://doi.org/10.1108/JHLSCM-11-2020-0108
- [66] Ma, Y., Xu, W., Qin, L., Zhao, X., & Du, J. (2019). Emergency shelters location-allocation problem concerning uncertainty and limited resources: a multi-objective optimization with a case study in the Central area of Beijing, China. *Geomatics, Natural Hazards and Risk, 10*(1), 1242–1266. <u>https://doi.org/10.1080/19475705.2019.1570977</u>
- [67] Chen, J., Hu, M., Shen, H., Lan, H., & Wu, Z. (2020). Study of Modeling Earthquake Emergency Rescue Material Scheduling Problems by Multi-objective Optimization Algorithms. 2020 2nd International Conference on Industrial Artificial Intelligence (IAI), 1–5. https://doi.org/10.1109/IAI50351.2020.9262178
- [68] Fu, L., Ding, M., & Zhang, Q. (2022). Flood risk assessment of urban cultural heritage based on PSR conceptual model with game theory and cloud model – A case study of Nanjing. *Journal of Cultural Heritage*, 58, 1–11. <u>https://doi.org/10.1016/j.culher.2022.09.017</u>
- [69] Cao, M., Tian, L., & Li, C. (2020). A Secure Video Steganography Based on the Intra-Prediction Mode (IPM) for H264. Sensors, 20(18), 5242. <u>https://doi.org/10.3390/s20185242</u>
- [70] Torresan, S., Furlan, E., Critto, A., Michetti, M., & Marcomini, A. (2020). Egypt's Coastal Vulnerability to Sea-Level Rise and Storm Surge: Present and Future Conditions. *Integrated Environmental Assessment and Management*, 16(5), 761–772. https://doi.org/10.1002/ieam.4280
- [71] Mohamed, S. A. (2020). Coastal vulnerability assessment using GIS-Based multicriteria analysis of Alexandrianorthwestern Nile Delta, Egypt. Journal of African Earth Sciences, 163, 103751. <u>https://doi.org/10.1016/j.jafrearsci.2020.103751</u>
- [72] EL-GEZIRY, T. M. (2020). On the Vulnerability of the Egyptian Mediterranean Coast to the Sea Level Rise. *Athens Journal of Sciences*, 7(4), 195–206. https://doi.org/10.30958/ajs.7-4-1
- [73] El-Masry, E. A., El-Sayed, M. K., Awad, M. A., El-Sammak, A. A., & Sabarouti, M. A. El. (2022). Vulnerability of tourism to climate change on the Mediterranean coastal area of El Hammam–EL Alamein, Egypt. *Environment, Development and Sustainability*, 24(1), 1145–1165. <u>https://doi.org/10.1007/s10668-021-01488-9</u>
- [74] Henderi, H. (2021). Comparison of Min-Max normalization and Z-Score Normalization in the K-nearest neighbor (kNN) Algorithm to Test the Accuracy of Types of Breast Cancer. *IJIIS: International Journal of Informatics and Information Systems*, 4(1), 13–20. https://doi.org/10.47738/ijiis.v4i1.73
- [75] Kalita, I., & Roy, M. (2022). Inception time DCNN for land cover classification by analyzing multi-temporal remotely sensed images. *IGARSS 2022 - 2022 IEEE International Geoscience and Remote Sensing Symposium*, 2022-July, 5736–5739. https://doi.org/10.1109/IGARSS46834.2022.9884128
- [76] Cantero-Chinchilla, S., Simpson, C. A., Ballisat, A., Croxford, A. J., & Wilcox, P. D. (2023). Convolutional neural networks for ultrasound corrosion profile time series regression. NDT & E International, 133, 102756. <u>https://doi.org/10.1016/j.ndteint.2022.102756</u>
- [77] Ismail Fawaz, H., Lucas, B., Forestier, G., Pelletier, C., Schmidt, D. F., Weber, J., Webb, G. I., Idoumghar, L., Muller, P.-A., & Petitjean, F. (2020). InceptionTime: Finding AlexNet for time series classification. *Data Mining and Knowledge Discovery*, 34(6), 1936–1962. <u>https://doi.org/10.1007/s10618-020-00710-y</u>
- [78] Wang, J., & Feng, L. (2016). Curve-fitting models for fossil fuel production forecasting: Key influence factors. Journal of Natural Gas Science and Engineering, 32, 138–149. https://doi.org/10.1016/j.jngse.2016.04.013. https://doi.org/10.1016/j.jngse.2016.04.013
- [79] Faisal, A. Al, Kafy, A.- Al, Afroz, F., & Rahaman, Z. A. (2023). Exploring and forecasting spatial and temporal patterns of fire hazard risk in Nepal's tiger conservation zones. *Ecological Modelling*, 476, 110244. <u>https://doi.org/10.1016/j.ecolmodel.2022.110244</u>

- [80] Mohamed, M., & El-Saber, N. (2023). Toward Energy Transformation: Intelligent Decision-Making Model Based on Uncertainty Neutrosophic Theory. *Neutrosophic Systems with Applications*, 9(SE-Articles), 13–23. <u>https://doi.org/10.61356/j.nswa.2023.65</u>
- [81] Abdel-Basset, M., Mohamed, M., Abdel-monem, A., & Elfattah, M. A. (2022). New extension of ordinal priority approach for multiple attribute decision-making problems: design and analysis. *Complex & Intelligent Systems*, 8(6), 4955–4970. https://doi.org/10.1007/s40747-022-00721-w
- [82] Andreas, A., Mavromoustakis, C. X., Mastorakis, G., Mumtaz, S., Batalla, J. M., & Pallis, E. (2020). Modified Machine Learning Techique for Curve Fitting on Regression Models for COVID-19 projections. 2020 IEEE 25th International Workshop on Computer Aided Modeling and Design of Communication Links and Networks (CAMAD), 2020-Septe, 1–6. <u>https://doi.org/10.1109/CAMAD50429.2020.9209264</u>
- [83] Karunasingha, D. S. K. (2022). Root mean square error or mean absolute error? Use their ratio as well. Information Sciences, 585, 609–629. https://doi.org/10.1016/j.ins.2021.11.036
- [84] Mutavhatsindi, T., Sigauke, C., & Mbuvha, R. (2020). Forecasting Hourly Global Horizontal Solar Irradiance in South Africa Using Machine Learning Models. *IEEE Access*, 8, 198872–198885. <u>https://doi.org/10.1109/ACCESS.2020.3034690</u>
- [85] Bai, S.-B., Wang, J., Lü, G.-N., Zhou, P.-G., Hou, S.-S., & Xu, S.-N. (2010). GIS-based logistic regression for landslide susceptibility mapping of the Zhongxian segment in the Three Gorges area, China. *Geomorphology*, 115(1–2), 23–31. https://doi.org/10.1016/j.geomorph.2009.09.025